Author's response to reviews

Title: Evaluation of Stepping Stones as a tool for changing knowledge, attitudes and behaviors associated with gender, relationships and HIV risk in Karnataka, India

Authors:

Janet Bradley (jbradley@uresp.ulaval.ca)
Parinita Bhattacharjee (parinita@khpt.org)
Meghna Girish (megrish2001@gmail.com)
BM Ramesh (bmramesh@khpt.org)
Arup Kumar Das (arup@khpt.org)

Version: 2 Date: 28 July 2010

Author's response to reviews:

Dear Sir/Madam

I would like to submit the attached manuscript, for consideration in BMC Public Health entitled:

Evaluation of Stepping Stones as a tool for changing knowledge, attitude behaviour, improving knowledge, changing attitudes and creating behaviour change essential for reducing STIs, HIV and gender based violence in Karnataka, India

This content of this work has not been published elsewhere, nor is simultaneously being considered for publication elsewhere.

We feel that this work provides a much needed evaluation of the wider community impact of the HIV intervention, Stepping Stones, and will be an excellent fit for your journal.